\documentclass[12pt]{article}

%\usepackage{enumerate}
\usepackage{calculus}
\usepackage[dvips]{graphics}

\pagestyle{empty}
\setlength{\parindent}{0pt}
\everymath{\displaystyle}
\newcommand{\answerspace}{\vspace{1.0in}}
\newcommand{\problemspace}{\bigskip}


\begin{document}

\textbf{Math 225: Calculus III}
\hfill\makebox[2.5in][r]{Name:\hrulefill}

\textbf{Quiz 1}\quad \textsl{Jan. 24/26, 1995}
\hfill\makebox[2.5in][r]{Section:\hrulefill}

\begin{enumerate}
\problemspace
\item 
Find the interior angles, $\theta$, $\phi$, of the parallelogram with vertices
$(0,0)$, $(2,1)$, $(3,4)$, and $(5,5)$.

\begin{center}

\resizebox{!}{1.25in}{\includegraphics{quiz1.eps}}

\end{center}
\answerspace

\problemspace
\item 
Find a vector that is perpendicular to both of the vectors $\i + \j +
3\k$ and $2\i + \j + \k$.

\answerspace
\end{enumerate}

\end{document}
